Author’s response to reviews

Title: Dementia and cognitive disorder identified at a forensic psychiatric examination - a study from Sweden

Authors:

Anette Ekström (Anette.Ekstrom@ptj.se)
Marianne Kristiansson (Marianne.Kristiansson@rmv.se)
Karin Björkstén (Karin.Sparring.Bjorksten@ki.se)

Version: 4 Date: 01 Sep 2017

Author’s response to reviews:

Anne Menard
BMC Geriatrics

Answers to your last (?) comments

1. A Conclusion heading is inserted above the conclusion section.

2. Availability of data and materials now reads:

“The datasets used and analysed during the current study are available from the corresponding author on reasonable request.”

3. “Ethics approval and consent to participate” is changed to “Ethical approval and consent to participate”
4. The paper has been cleaned by using “accept all changes” in the word programme.

Karin Sparring Björkstén